Author's response to reviews

Title: Community Satisfaction with the Urban Health Extension Service in South Ethiopia and Associated Factors

Authors:

Ephrem Lejore (ephrlej@gmail.com)
Tezera Moshago (moshagot1@gmail.com)

Version: 2 Date: 24 May 2014

Author's response to reviews:

Dear BioMed Central Editorial Team

Firstly, we would like to appreciate your skills and comments. Based on your requirements, we seriously went through all points where you emphasized your comments and consequently, we included all what you enquired. We thank your consideration and all goodesses to this consideration.

Abstract maintained to the limited word and recommended structure
The body of the paper revised as per the guideline
- Competing interests
- Authors' contributions
- Authors' information added to the body of the paper